Author's response to reviews

Title: Mindboggle: Automated brain labeling with multiple atlases

Authors:

Arno Klein (arno@binarybottle.com)
Brett Mensh (brett@mensh.com)
Satrajit Ghosh (satra@speech.mit.edu)
Jason Tourville (jtour@bu.edu)
Joy Hirsch (jh2155@columbia.edu)

Version: 4 Date: 25 August 2005

Author’s response to reviews:

As per the reviewer's request, I have removed all references to linear labeling in the text and the figures. The manuscript now deals solely with nonlinear labeling using Mindboggle, and presents evidence of increased accuracy in anatomical labeling of brain image data when using multiple atlases. I have also clarified portions of the text, including the description of Mindboggle in the Methods section and have added two related figures: a flowchart outlining Mindboggle and a figure portraying Mindboggle steps prior to labeling.